

BANK OF ENGLAND

Staff Working Paper No. 783 The real effects of zombie lending in Europe Belinda Tracey

May 2021

This is an updated version of the Staff Working Paper originally published on 1 March 2019

Staff Working Papers describe research in progress by the author(s) and are published to elicit comments and to further debate. Any views expressed are solely those of the author(s) and so cannot be taken to represent those of the Bank of England or to state Bank of England policy. This paper should therefore not be reported as representing the views of the Bank of England or members of the Monetary Policy Committee, Financial Policy Committee or Prudential Regulation Committee.



BANK OF ENGLAND

Staff Working Paper No. 783 The real effects of zombie lending in Europe Belinda Tracey⁽¹⁾

Abstract

'Zombie lending' occurs when a lender supports an otherwise insolvent borrower. Recent studies document that zombie lending to European firms has been widespread following the onset of the European sovereign debt crisis. This paper develops a quantitative model to study the impact of these lending practices on the dynamics and financial decisions of firms. In the model, firm liquidations and zombie lending arise endogenously. The model provides a good match to key euro-area firm statistics over the period 2011 to 2014. A key finding is that zombie lending has a substantial impact on borrowing costs, helping more low-productivity firms to survive. This, in turn, causes a drag on aggregate output, investment and productivity. These results suggest that zombie lending practices contributed to the lower output experienced by the euro area following the onset of the sovereign debt crisis.

Key words: Zombie lending, forbearance lending, zombie firms, firm defaults, firm dynamics.

JEL classification: G32, G33, L25.

(1) Bank of England and CfM. Email: belinda.tracey@bankofengland.co.uk

I would like to acknowledge the use of the University of Oxford Advanced Research Computing facility in carrying out this work. http://dx.doi.org/10.5281/zenodo.22558. I wish to thank Wouter Den Haan, Andrea Ferrero, Basile Grassi, John Muellbauer, Wojtek Paczos, Carlo Pizzinelli, Isabel Schnabel and Francesco Zanetti for detailed comments and conversations, as well as participants at the Oxford Macroeconomics Working Group Seminar Series, 2019 Royal Economic Society Annual Conference, 2019 Workshop on Economic Science with Heterogeneous Interacting Agents, 2019 Computing in Economics and Finance Conference, and 2019 Annual Congress of the European Economic Association.

The views expressed in this paper are those of the author, and not necessarily those of the Bank of England or its committees.

The Bank's working paper series can be found at www.bankofengland.co.uk/working-paper/staff-working-papers

Bank of England, Threadneedle Street, London, EC2R 8AH Email enquiries@bankofengland.co.uk

© Bank of England 2021 ISSN 1749-9135 (on-line)

1 Introduction

In the late 1990s, around 30 per cent of Japanese firms were in receipt of forbearance lending - or so-called "zombie lending" (Caballero, Hoshi and Kashyap, 2008). These lending practices came to the fore following the 1997 financial crisis in Japan, and were accompanied by a weakened banking sector and a slow recovery. Some argue that forbearance lending was not only a by-product of economic conditions, but in fact contributed to the low output growth experienced by Japan during their lost decade (see, e.g., Peek and Rosengren, 2005, Caballero, Hoshi and Kashyap, 2008).

Fast forward almost 20 years and there are some notable similarities between the Japanese experience and the euro area following the onset of the European sovereign debt crisis and COVID-19 pandemic. Recent studies have uncovered widespread forbearance lending to European firms (see, e.g., Acharya et al., 2019, Banerjee and Hofmann, 2020, Blattner, Farinha and Rebelo, 2018, Adalet McGowan, Andrews and Millot, 2018, Schivardi, Sette and Tabellini, 2017). Weak banks have extended credit to weak firms in order to avoid the declaration of nonperforming loans on their own balance sheets (see, e.g., Acharya et al., 2019, Acharya et al., 2020*b*, Andrews and Petroulakis, 2019, Blattner, Farinha and Rebelo, 2018, Schivardi, Sette and Tabellini, 2017, Storz et al., 2017). In this paper, I study the impact of these lending practices on the financial decisions of firms, firm dynamics and economic performance. To do so, my main contribution is the development of a quantitative model that incorporates a role for forbearance lending.

The vehicle of analysis is a dynamic model of heterogeneous firms in which: (i) firms can obtain a loan from a lender to be used for production and they also have the option to default in each period; (ii) if a firm defaults, it has the option to either liquidate or to obtain loan forbearance; (iii) loan forbearance takes the form of renegotiation over an *ex post* reduction to the outstanding loan repayment, where the extent of the reduction is determined endogenously in a Nash bargaining mechanism; and (iv) lenders face information asymmetry because they observe the overall firm productivity draw but do not know precisely whether the draw belongs to a "low-quality" or "high-quality" firm.

I focus on the euro area after the onset of the European sovereign debt crisis that commenced in 2009, a period marked by high levels of forbearance lending levels. As many as 10 per cent of European firms were in receipt of subsidized bank loans in 2014 (Acharya et al., 2019). As such, the model is calibrated using euro area firmlevel statistics over the period 2011 to 2014, primarily obtained from the Amadeus Database. The calibrated model provides a good match to key firm-level statistics, including the prevalence of forbearance lending and firm defaults.

The quantitative results from the calibrated model provide insights about the characteristics and dynamics of firms in receipt of forbearance lending. Compared to their non-zombie counterparts, these firms have lower sales growth, lower productivity, higher leverage and are smaller in scale. Firms are susceptible to becoming zombies when faced with a sequence of bad productivity shocks; these firms reduce their scale in order to avoid costly equity issuance, and increase their leverage, thereby pushing up their risk of loan forbearance. These results are consistent with the empirical findings of Banerjee and Hofmann (2020).

To evaluate the quantitative impact of forbearance lending, I conduct a counterfactual exercise in which firms have the option to default but no longer have access to any form of loan forbearance. The results of my analysis can be summarized as follows. First, I show that the rate of firm liquidation is only a little higher in the counterfactual scenario with no forbearance lending as compared to the benchmark scenario with forbearance lending. This is because firms behave differently; in the absence of forbearance lending, the risk of liquidation increases because liquidation is now the only default option. In turn, borrowing costs increase, as lenders take account of this increased risk of liquidation in their loan pricing. But firms respond by taking on significantly less leverage, which results in a counterfactual liquidation rate that is little changed from the benchmark scenario.

Second, I find that the averages of firms' growth, investment rates and total factor productivity are higher in the counterfactual scenario with no forbearance lending. A key driver of this result is that there is a larger proportion of high-quality firms in the counterfactual scenario; the forbearance lending that is present in the benchmark scenario helps to prevent the creative destruction of low-quality firms. The combination of these effects results in an increase in aggregate output, investment and total factor productivity in the counterfactual scenario with no forbearance lending. These model results complement several empirical studies that find zombie firm survival can impair aggregate productivity growth and the pace of recovery following recessions (see, e.g., Andrews and Petroulakis, 2019, Blattner, Farinha and Rebelo, 2018, Caballero, Hoshi and Kashyap, 2008, Jorda et al. (2020), Adalet McGowan, Andrews and Millot 2017, Adalet McGowan, Andrews and Millot, 2018). Taken together, my results suggest that forbearance lending practices contributed to the lower output experienced by the euro area following the onset of the European sovereign debt crisis. I also find that the vast majority of firms in receipt of forbearance lending in the benchmark scenario are low-quality. Several studies show that lower quality firms are more likely to receive some kind of loan forbearance because lenders face zombie lending incentives in order to avoid taking a balance sheet hit due to the write-off of bad loans (see, e.g., Acharya et al., 2019, Andrews and Petroulakis, 2019, Blattner, Farinha and Rebelo, 2018, Giannetti and Simonov, 2013, Schivardi, Sette and Tabellini, 2017, Storz et al., 2017). My model highlights that information asymmetry faced by lenders is another key driver. In the model, lenders do take account of firm-type risk. But because they cannot perfectly predict firm quality, borrowing costs are cheaper than otherwise for a low-quality firm, and vice versa for a high-quality firm. The end result is that credit is (mis-)allocated to low-quality firms that have higher forbearance risks.

It is important to note that the model does not incorporate a role for "zombie lending incentives" by weak banks. Rather, I outline a framework that allows for an evaluation of how zombie lending affects the dynamics of firms. In this respect, I build on existing empirical research that has focused on zombie lending incentives and the consequences of forbearance lending (see, e.g., Acharya et al., 2019, Acharya et al., 2020*a*, Acharya et al., 2020*b*, Andrews and Petroulakis, 2019, Blattner, Farinha and Rebelo, 2018, Bonfim et al., 2020, Caballero, Hoshi and Kashyap, 2008, Schivardi, Sette and Tabellini, 2017, Storz et al., 2017). Compared to those studies, which are largely based on reduced form analyses, this paper provides new insights about the impact of forbearance lending on firm behavior and their dynamics. Moreover, the structural approach facilitates counterfactual analyses to determine the impact of these lending practices.

Additionally, this paper contributes to the literature on factor misallocation, financial frictions and firm dynamics (see, e.g. Bassetto, Cagetti and De Nardi, 2015, Gopinath et al., 2017, Hsieh and Klenow, 2009, Khan and Thomas, 2013, Midrigan and Xu, 2014, Moll, 2014, Restuccia and Rogerson, 2008), which examines how differences in the efficiency of factor allocation can explain observed differences in total factor productivity and output. In these studies, financial frictions typically take the form of incentive compatibility or collateral constraints. I document new evidence about misallocation that arises due to financial frictions as well as endogenous borrowing costs generated by equilibrium default.

The paper also relates to several studies that attempt to model the influence of forbearance lending in Japan. To be more precise, Caballero, Hoshi and Kashyap (2008) outline a stylized model of entry and exit to consider the impact of zombies on firm creation and productivity. Kwon, Narita and Narita (2015), in turn, estimate the impact of zombie lending on aggregate output via a counterfactual analysis that applies the distribution of factor-input wedges estimated for non-zombies to all firms. The present study extends this research by incorporating an explicit role for firm dynamics, loan financing and firm liquidation; these are key factors that are likely to influence any analysis of forbearance lending.

Finally, the paper adds to a growing body of literature that incorporates a role for multiple firm default choices within a dynamic model of heterogeneous firms. In this context, the firm default options in my model are similar to those of Arellano, Bai and Zhang (2012), Corbae and D'Erasmo (2021), Senkal (2014) and Tamayo (2017). The firm equilibrium model in this study, as well as in Corbae and D'Erasmo (2021) and Senkal (2014), builds on other corporate finance models (see, e.g., Cooley and Quadrini, 2001, Hennessy and Whited, 2007), by incorporating endogenous entry similar to that of Hopenhayn and Rogerson (1993) as well as endogenous renegotiation similar to that of Yue (2010). Aside from the fact that I employ the model to study forbearance lending rather than financial development (see, e.g., Arellano, Bai and Zhang, 2012) or U.S. bankruptcy laws (see e.g., Corbae and D'Erasmo (2021), Senkal, 2014 and Tamayo, 2017), another key point of departure from these previous studies is that the model includes a role for information asymmetry about firm-type.¹

The remainder of this paper is structured as follows. Section 2 outlines the model environment. Section 3 describes the recursive equilibrium of the model. Section 4 discusses the model calibration and analyzes the model equilibrium. Section 5 quantitatively evaluates the impact of forbearance lending on firm dynamics and performance. Section 6 concludes.

2 The Model Economy

2.1 Firms and technology

This section sets out a discrete-time model to study the impact of forbearance lending, which builds on several other firm equilibrium models.² In the model, entrepreneurs are infinitely lived. They have access to a pool of risky projects of mass

¹The model of Tamayo (2017) features a different type of information asymmetry, with the lender being unable to observe any form of firm productivity but having the option to engage in various types of monitoring.

²Namely, the model and default options are comparable to those of Arellano, Bai and Zhang (2012), Corbae and D'Erasmo (2021), Senkal (2014) and Tamayo (2017).

one, which are referred to as firms. Each entrepreneur owns at most one firm. An entrepreneur that owns firm j, chooses physical capital k_{jt+1} and a new one period loan contract $(l_{jt+1}, l_{R,jt+1})$ to maximize the expected present value of all current and future dividends:

$$E_0 \sum_{t=0}^{\infty} \beta^t \left[d_{jt} \left(1 + \lambda I_{\{d_{jt} < 0\}} \right) \right], \tag{1}$$

where $\beta \in (0, 1)$ is the discount factor of all firms, λ is a proportional cost of equity issuance, and d_{jt} is the dividend function given by:

$$d_{jt} = y_{jt} - k_{jt+1} + (1 - \delta) k_{jt} - l_{R,jt} + l_{jt+1} - \frac{\phi i_{jt}^2}{2k_{jt}} - \chi_c$$
(2)

In Equation (2), y_{jt} is output produced by firm j in period $t, \delta \in (0, 1)$ is the depreciation rate for physical capital, $k_{jt+1} - (1 - \delta) k_{jt}$ is investment $i_{jt}, l_{R,jt}$ is the repayment on the previous period's loan $l_{jt}, \phi i_{jt}^2/(2k_{jt})$ are capital adjustment costs, and χ_c is a fixed cost of operation that firms must pay to produce. The price of output is normalized to one. Firms produce output according to a decreasing returns to scale production technology $y_{jt} = z_{jt}k_{jt}^{\alpha}$, where z_{jt} is an idiosyncratic productivity process. The probability distribution of firm j's productivity z_{jt} is conditional on the previous realization z_{jt-1} and follows a Markov process given by $f(z_{jt}, z_{jt-1})$.

I assume that firm productivity is a log-normal AR(1) process: $\ln(z_{jt}) = \mu_j (1 - \varphi) + \varphi \ln(z_{jt-1}) + \epsilon$, with $|\varphi| < 1$ and $\epsilon \sim \mathcal{N}(0, \sigma_{\epsilon}^2)$. There are two firm types, which differ according to the permanent productivity parameter μ_j . "Low-quality" firms have a permanent productivity μ_b and "high-quality" firms have a permanent productivity μ_g , such that $\mu_b < \mu_g$.

An idle entrepreneur will start a new firm if he receives a project opportunity. Entrants start with zero capital and debt, and decide on the future optimal capital and a future loan before they know their future productivity. Their future productivity is drawn from the stationary probability distribution $g_j(z_{jt+1})$ derived from $f_j(z_{jt+1}, z_{jt})$, for which the permanent productivity μ_j is drawn from a Bernoulli distribution with $\Pr(\mu_j = \mu_b) = 0.5$. The mass of all risky projects available to both entrants and operating firms is always one. As such, a new risky project becomes available for a new entrepreneur to start a firm when an operating entrepreneur decides to liquidate its firm.

2.2 Lenders and model timing

Competitive lenders face information asymmetry when they initially make a loan because they observe z_{jt} but not μ_j . This implies that lenders do not know with certainty whether a firm is low-quality or high-quality, and this is referred to as "firm-type risk". Lenders measure firm-type risk based on their assumed knowledge of the proportion of firms that are low-quality versus high-quality combined with information about the conditional distributions of z_{jt} given μ_j . A full description of this information asymmetry is provided in Section 3.5.

Lenders are risk-neutral and can borrow or lend as much as is needed from international capital markets at a constant interest rate r > 0. I denote $\Omega(k_{jt+1}, z_{jt})$ as the set of firm-specific loan schedules available to a firm with next period capital k_{jt+1} and productivity z_{jt} ; each contract $(l_{jt+1}, l_{R,jt+1}) \in \Omega(k_{jt+1}, z_{jt})$ maps a current one-period loan l_{jt+1} to a next period repayment amount $l_{R,jt+1}$.

Entrepreneurs have the option to default on the loan amount owing in each period. At the beginning of each period t, the model timing and entrepreneur options are:

- 1. Productivity z_{jt} is realized, and the state space for incumbent firm j is $\{k_{jt}, l_{R,jt}, z_{jt}, \mu_j\}$.
- 2. Entrepreneurs choose from the following options for their incumbent firm j, which includes two default options:
 - (a) **Continuation**. Firm j continues operating and repays the full loan amount $l_{R,jt}$. The entrepreneur chooses physical capital k_{jt+1} and a new loan contract $(l_{jt+1}, l_{R,jt+1})$.
 - (b) **Liquidation**. Firm *j* defaults on the full loan amount owing, $l_{R,jt}$, and liquidates its assets at a firesale discount $\chi_d < 1$ where $\chi_d k_{jt}$ is the firesale firm recovery value. The entrepreneur receives the liquidation value $\max{\{\chi_d k_{jt} l_{R,jt}, 0\}}$; the lender receives $\min{\{\chi_d k_{jt}, l_{R,jt}\}}$.
 - (c) Loan forbearance. Firm j obtains loan forbearance, which takes the form of post-default renegotiation over the outstanding loan amount, $l_{R,jt}$. The entrepreneur bargains with the lender over the loan repayment fraction $\psi(k_{jt}, l_{R,jt}, z_{jt}, \mu_j)$ in a Nash bargaining mechanism. The repayment $\psi(k_{jt}, l_{R,jt}, z_{jt}, \mu_j)$ is restricted to the interval [0, 1]. When the entrepreneur and the lender agree on the repayment fraction ψ , the entrepreneur repays the reduced loan amount $\psi l_{R,jt}$ in addition to a renegotiating cost proportional to its capital stock $\chi_f k_{jt}$ in the default period.

The entrepreneur continues to operate, chooses physical capital k_{jt+1} , and has access to new loan finance in the default period.

3. Entrance. Idle entrepreneurs make an entry decision about whether to start a firm or not. Their initial productivity is drawn from the stationary distribution $g(z_{jt+1})$ derived from $f(z_{jt+1}, z_{jt})$, for which the permanent productivity μ_j is drawn from a Bernoulli distribution with $\Pr(\mu_j = \mu_b) = 0.5$.

Lenders offer a firm-specific loan schedule $\Omega(k_{jt+1}, z_{jt})$ such that they break even in expected value on each loan. The loan schedule incorporates firm-specific liquidation risk, forbearance risk and firm-type risk. The lender's problem is outlined formally in Section 3.5.

3 Recursive Equilibrium

In this section, I define a stationary recursive equilibrium for the model. For all variables, I drop the time and firm subscripts for ease of exposition and employ the notation: $x_{jt-1} = x^-$, $x_{jt} = x$ and $x_{jt+1} = x'$. Given the states $s = (k, l_R, z, \mu)$, the equilibrium is determined by the entrepreneur's policy rules for capital k'(s), for loans $l'_R(s)$, the loan schedule $\Omega(k', z)$, and the default policy. And given any loan schedule, $\Omega(k', z)$, and loan recovery schedule, $\psi(s)$, the entrepreneur solves its firm optimization problem with perfect information about all states s. The lender, however, does not observe μ when initially issuing a loan to an entrepreneur.

3.1 The Firm's Problem

The entrepreneur chooses whether to liquidate the firm, receive loan forbearance or to repay the loan in order to maximize the present value of current and future profits, given the productivity shock z, the permanent productivity μ , the initial capital kand the loan repayment amount l_R . The corresponding value function of the firm is:

$$V(k, l_R, z, \mu) = \max_{\{c, d, f\}} \{ V_c(k, l_R, z, \mu), V_d(k, l_R), V_f(k, l_R, z, \mu) \},$$
(3)

where $V_c(k, l_R, z, \mu)$ is the value function if the entrepreneur does not default and continues to operate; $V_d(k, l_R)$ is the value function if the entrepreneur defaults and liquidates the firm; and $V_f(k, l_R, z, \mu)$ is the value function for forbearance lending when the entrepreneur defaults and renegotiates with the lender about the repayment fraction of the original loan. The decision to default or repay the loan is a period-by-period decision. The value function conditional on the firm not defaulting and continuing to operate is:

$$V_{c}(k, l_{R}, z, \mu) = \max_{k', (l', l'_{R}) \in \Omega} d_{c} \left(1 + \lambda I_{\{d_{c} < 0\}} \right) + \beta \int_{z'} V(k', l'_{R}, z', \mu) f(z', z) dz',$$
(4)

where $d_c = zk^{\alpha} - k' + (1 - \delta) k - l_R + l' - \frac{\phi i^2}{2k} - \chi_c$ are the dividends for a continuing firm. Here the entrepreneur chooses the new level of optimal k' and a new loan contract (l', l'_R) to maximize the present value of current and future dividends for the firm. The loan schedule offered for new borrowing $\Omega(k', z)$ depends on the firm's choice of k' and on the state z but not on permanent productivity μ .

When the entrepreneur defaults on the loan and liquidates the firm, the value function is:

$$V_d(k, l_R) = \max\{\chi_d k - l_R, 0\}.$$
 (5)

If the entrepreneur decides to default and obtain loan forbearance, the firm must repay a reduced fraction $\psi(k, l_R, z, \mu)$ of the unpaid loan repayment amount l_R . The recovery rate $\psi(k, l_R, z, \mu)$ is determined endogenously in a Nash bargaining mechanism explained next in Section 3.4. In the period of default, the reduced loan is repaid and the firm continues to have access to new loan finance. This continued access to credit is akin to "evergreening", whereby further credit is extended to a troubled firm. The value function associated with a firm that defaults but repays the agreed reduced fraction of the outstanding loan is:

$$V_{f}(k, l_{R}, z, \mu) = \max_{\substack{k', (l', l'_{R}) \in \Omega}} d_{f} \left(1 + \lambda I_{\{d_{f} < 0\}} \right) +\beta \int_{z'} V(k', l'_{R}, z', \mu) f(z', z) dz',$$
(6)

where $d_f = zk^{\alpha} - k' + (1 - \delta)k - \psi(k, l_R, z, \mu)l_R + l' - \frac{\phi i^2}{2k} - \chi_c - \chi_f k$ are the dividends for a firm with forbearance lending, and ψ is the fraction of the unpaid loan repayment amount l_R .

3.2 Default Policies

An entrepreneur's default policy can be characterized by a continuing set $C(k, l_R, \mu)$, a liquidation set $D(k, l_R, \mu)$ and a forbearance lending set $F(k, l_R, \mu)$. These three sets of z's are mutually exclusive and specify when a particular default option is optimal for a given level of capital k and loan contract (l, l_R) . They are respectively defined as:

$$C(k, l_R, \mu) = \{ z \in Z : V_c(k, l_R, z, \mu) \ge \max\{ V_d(k, l_R), V_f(k, l_R, z, \mu) \} \},$$
(7)

$$D(k, l_R, \mu) = \{ z \in Z : V_d(k, l_R) > \max\{V_c(k, l_R, z, \mu), V_f(k, l_R, z, \mu)\} \},$$
(8)

$$F(k, l_R, \mu) = \{ z \in Z : V_f(k, l_R, z, \mu) > \max\{V_c(k, l_R, z, \mu), V_d(k, l_R)\} \}.$$
(9)

3.3 Entrants

When an idle entrepreneur receives a project opportunity, he will attempt to start a new firm. Entrants choose their optimal capital and loan before they know their future productivity or type. Entrants' future productivity is drawn from the stationary distribution g(z') derived from f(z', z), for which permanent productivity μ is drawn from a Bernoulli distribution with $Pr(\mu = \mu_b) = 0.5$. The value function associated with a potential entrant is given by:

$$V_{e} = \max_{k', (l', l'_{R}) \in \Omega} d_{e} \left(1 + \lambda I_{\{d_{c} < 0\}} \right) + \int_{z'} V\left(k', l'_{R}, z', \mu\right) g\left(z'\right) dz', \tag{10}$$

where $d_e = -k' + l'$, and I assume that the equity issuance costs are the same as for continuing firms.

3.4 The Forbearance Problem

When a firm chooses the option to obtain loan forbearance, this default option takes the form of a bargaining game regarding the fraction $\psi(k, l_R, z, \mu)$ of the outstanding loan repayment l_R . Here the value of the defaulted loan repayment amount l_R is reduced to $\psi(k, l_R, z, \mu) l_R$. The value of such an agreement to the firm is equivalent to the present value of all future expected profits of forbearance when the loan recovery rate is $\psi(k, l_R, z, \mu)$, as described by Equation (6). The lender gets the value of the reduced loan repayment $\psi(k, l_R, z, \mu) l_R$.

The threat point of the Nash bargaining mechanism is firm liquidation, whereby the entrepreneur receives the liquidation value $V_d(k, l_R, z, \mu)$ and the lender receives the firesale recovery value min $\{\chi_d k, l_R\}$.

I let Δ^f $(p; k, l_R, z, \mu)$ denote the firm's surplus in the Nash bargaining agreement, which is the difference between the value of accepting the loan recovery rate $p = \psi(k, l_R, z, \mu)$ and the value of rejecting it, given the firm's capital k, the loan repayment amount l_R , the firm's productivity z, and the firm permanent productivity μ . The firm's surplus is:

$$\Delta^{f}(p;k,l_{R},z,\mu) = V_{f}(p;k,l_{R},z,\mu) - V_{d}(k,l_{R}).$$
(11)

The firm surplus will differ for a high-quality firm $(\mu = \mu_g)$ and a low-quality firm $(\mu = \mu_b)$. While the lender faces information asymmetry about firm-type when initially issuing a loan, firm-type is revealed during renegotiation from the firm's surplus. I am therefore using a Nash bargaining mechanism with perfect information to obtain the solution for the repayment fraction $\psi(k, l_R, z, \mu)$.³

I let $\Delta^b(p; k, l_R)$ denote the risk-neutral lender's surplus in the bargaining agreement, which is the present vale of the recovered loan repayment. The lender's surplus is:

$$\Delta^b (p; k, l_R) = p l_R - \min\left\{\chi_d k, l_R\right\}$$
(12)

The lender can extract loan repayments up to the full amount of a firm's cost of default when it has all of the bargaining power. Alternatively, the firm can obtain the maximum reduction of the loan, which is the difference between l_R and the recovery value for the lender, when it has all of the bargaining power. I consider the general case and assume that the borrower has bargaining power θ and the lender has bargaining power $(1 - \theta)$. For any capital stock k, loan repayment amount l_R , and productivity shock z, I define the bargaining power set as $\Theta \subset [0, 1]$ for $\theta \in \Theta$ in order to ensure that the renegotiation surplus has a unique optimum.

Given the capital stock k, loan repayment l_R , and productivity shock z: the loan recovery rate $\psi(k, l_R, z, \mu)$ solves the following bargaining problem:

$$\psi(k, l_R, z, \mu) = \arg \max_{p \in [0,1]} \left[\Delta^f(p; k, l_R, z, \mu) \right]^{\theta} \left[\Delta^b(p; k, l_R) \right]^{1-\theta}$$

$$\text{s.t.} \quad \Delta^f(p; k, l_R, z, \mu) \ge 0,$$

$$\Delta^b(p; k, l_R) \ge 0.$$
(13)

³That said, the Nash bargaining solution when there is perfect information should coincide with a Nash bargaining solution with incomplete information that maximizes the product of the expected surplus of the firm and the lender, given the lender's incomplete information, according to D'Erasmo (2011).

3.5 The Lender's Problem

When initially pricing and offering a loan schedule to a firm, I assume that the lender knows the liquidation risk and forbearance risk for all states but that it does not know μ . Thus, while the lender knows what the liquidation risk is for a low-quality firm, they do not know with certainty whether a given firm is low-quality or high-quality, and I refer to this as "firm-type risk". Lenders assess firm-type risk based on their assumed knowledge of the proportion of firms that are low-quality versus high-quality combined with information about the conditional distributions of z given μ .

More formally, a lender offers a loan schedule $\Omega(k', z)$ to maximize expected profits π_b , which are given by:

$$\pi_{b} (k', l'_{R}, z) = -l' + \sum_{j=b,g} \rho (\mu = \mu_{j} | z) \\ \times \left\{ \frac{1}{1+r} \left[1 - \rho_{f} (k', l'_{R}, z, \mu_{j}) - \rho_{d} (k', l'_{R}, z, \mu_{j}) \right] l'_{R} \\ + \frac{1}{1+r} \left(\rho_{f} (k', l'_{R}, z, \mu_{j}) \cdot \gamma (k', l'_{R}, z, \mu_{j}) \right) l'_{R} \\ + \frac{1}{1+r} \rho_{d} (k', l'_{R}, z, \mu_{j}) \min \left\{ \chi_{d} k', l'_{R} \right\} - \xi$$

$$(14)$$

where $\rho(\mu = \mu_j | z)$ is the probability that a firm has permanent productivity μ_j given it has observed productivity z (a measure of "firm-type risk"), $\rho_d(k', l'_R, z, \mu)$ is the expected probability of liquidation for a firm, $\rho_f(k', l'_R, z, \mu)$ is the expected probability of forbearance, and $\gamma(k', l'_R, z, \mu)$ is the expected recovery rate, given by the expected proportion of the loans that the creditors can recover, conditional on forbearance.

The lender must pay a fixed cost for each loan, ξ . The first term on the righthand side of Equation (14) represents the resources that the lender spends today. The remaining three terms on the right-hand side comprise the expected loan repayment amount l'_R , discounted by the risk-free rate and accounting for liquidation risk, forbearance risk, and firm-type risk.

The equilibrium loan schedule $\Omega(k', z)$ comprises all loan contracts (l', l'_R) that allow the lender to break even in expected values, accounting for the liquidation risk, the forbearance risk and the firm-type risk that the lender faces, and is described by:

$$l' = \sum_{j=b,g} \rho \left(\mu = \mu_j | z\right) \quad \left\{ \frac{1}{1+r} \left[1 - \rho_f \left(k', l'_R, z, \mu_j\right) - \rho_d \left(k', l'_R, z, \mu_j\right) \right] l'_R + \frac{1}{1+r} \left(\rho_f \left(k', l'_R, z, \mu_j\right) \cdot \gamma \left(k', l'_R, z, \mu_j\right) \right) l'_R + \frac{1}{1+r} \rho_d \left(k', l'_R, z, \mu_j\right) \min \left\{ \chi_d k', l'_R \right\} \right\} - \xi$$
(15)

where the probability of liquidation, ρ_d , the probability of forbearance, ρ_f , the expected recovery rate, γ , and the probability that a firm has permanent productivity μ_j given z, $\rho(\mu = \mu_j | z)$, are endogenous to the model. The effective interest rate for a loan contract (l', l'_R) is $r_L = l'_R/l' - 1$. The zero profit assumption and the fact that $0 \leq \rho_d \leq 1, 0 \leq \rho_f \leq 1$ and $0 \leq \gamma \leq 1$ imply that the gross firm effective interest rate lies in the interval $[(1 + r), \infty)$.

In equilibrium, the loan schedule $\Omega(k', z)$ must be consistent with the firm's optimization and with expected zero profits of the lender, such that the loan schedule correctly assesses the probability of liquidation, the probability of forbearance of firms, and the probability of firm-type.

The first term in Equation (15) shows that both liquidation risk ρ_d and forbearance risk ρ_f have a first-order effect on the loan schedule because they enter Equation (15) linearly. And the second term shows that the loan recovery rate γ affects the debt schedule through its combined effect with forbearance risk. Additionally, the loan recovery rate has an indirect impact on the *ex ante* liquidation risk, because it affects the second moment of the loan schedule.

The firm-type risk also effects the loan schedule. In comparison to a scenario of perfect information, high-quality firms will face a higher cost of borrowing and vice versa for lower quality firms. These differences in the borrowing costs, driven by the lenders inability to perfectly a firm's permanent productivity, will in turn influence the decision rules of firms.

3.5.1 Liquidation Risk

The liquidation probabilities $\rho_d(k', l'_R, z, \mu)$ are related to the liquidation sets $D(k', l'_R, \mu)$ as follows:

$$\rho_d(k', l'_R, z, \mu) = \int_{D(k', l'_R, \mu)} f(z', z) \, dz'.$$
(16)

This implies that when the liquidation sets are empty, $D(k', l'_R, \mu) = \emptyset$, the equilibrium liquidation probabilities $\rho_d(k', l'_R, z, \mu)$ equal zero. The liquidation probabilities $\rho_d(k', l'_R, z, \mu)$ equal one when $D(k', l'_R, \mu) = Z$.

3.5.2 Forbearance Risk

Similarly, the forbearance lending probabilities $\rho_f(k', l'_R, z, \mu)$ are related to the forbearance lending sets $F(k', l'_R, \mu)$ as follows:

$$\rho_f(k', l'_R, z, \mu) = \int_{F(k', l'_R, \mu)} f(z', z) \, dz'.$$
(17)

This similarly implies that when forbearance sets are empty, $F(k', l'_R, \mu) = \emptyset$, the equilibrium forbearance probabilities $\rho_f(k', l'_R, z, \mu)$ equal zero. The forbearance probabilities $\rho_f(k', l'_R, z, \mu)$ equal one when $F(k', l'_R, \mu) = Z$.

The expected recovery rate $\gamma(k', l'_R, z, \mu)$ is determined by:

$$\gamma(k', l'_R, z, \mu) = \frac{\int_{F(k', l'_R, \mu)} \psi(s') f(z', z) dz'}{\rho_f(k', l'_R, z, \mu)}$$
(18)

The numerator of Equation (18) is the proportion of the loan that the lender can expect to recover. The denominator of Equation (18) is the forbearance probability.

3.5.3 Firm-type Risk

I now consider the firm-type risk, which quantifies the information asymmetry faced by the lender. While lenders cannot observe μ , I assume they know: (a) the proportion of firms are low-quality versus high-quality, i.e. $\rho(\mu = \mu_b)$ and $\rho(\mu = \mu_g)$; and (b) they know the conditional distribution of z for a given μ .⁴ As such, the lender can determine the probability that a firm has permanent productivity μ_b given z, $\rho(\mu = \mu_b|z)$, by using Bayes' Rule:

$$\rho(\mu = \mu_b | z) = \frac{\rho(z | \mu = \mu_b) \rho(\mu = \mu_b)}{\rho(z | \mu = \mu_b) \rho(\mu = \mu_b) + \rho(z | \mu = \mu_g) \rho(\mu = \mu_g)}$$
(19)

where $\rho(\mu = \mu_b)$ in Equation (19) is the proportion of low-quality firms, given by the cross-sectional stationary distributions of firms $\Gamma(s)$ such that:

$$\rho\left(\mu=\mu_b\right) = \int \Gamma\left(k, l_R, z; \mu=\mu_b\right) d\left(k \times l_R \times z\right).$$
(20)

⁴I denote to the proportion of firms that are low quality, as determined by the cross-sectional distribution of firms, $\rho (\mu = \mu_b)$. This differs from the aforementioned initial probability that a firm is low-quality, $\Pr (\mu = \mu_b) = 0.5$, which is applies to new entrants.

In Equation (19), $\rho(z|\mu = \mu_b)$ is the probability that the observed z is from a log normal distribution with mean μ_b . The initial firm productivity process is distributed such that $\ln(z) \sim \mathcal{N}(\mu, \sigma_{\varphi}^2)$ for $\sigma_{\varphi}^2 = \frac{\sigma_{\epsilon}^2}{1 - \varphi^2}$. This initial process can be used to determine the initial conditional probabilities. But the conditional distribution for z is subsequently given by information from the cross-sectional stationary distributions of firms $\Gamma(s)$ and the assumption that the firm productivity process is log-normally distributed. The conditional distribution for z is approximated as follows. First, the mean of the firm productivity process for $\ln(z)$ given $\mu = \mu_b$ as:

$$E\left(\ln\left(z\right)|\mu=\mu_{b}\right) = \sum_{i} \ln\left(z_{i}\right) p_{i}$$
(21)

where the probability of a given z_i is $p_i = \frac{\int \Gamma(k, l_R, z_i, \mu = \mu_b) d(k \times l_R)}{\rho(\mu = \mu_b)}$.⁵ The variance of the process is:

$$Var(\ln(z) | \mu = \mu_b) = \sum_{i} (\ln(z_i) - E(\ln(z) | \mu = \mu_b))^2 p_i.$$
 (22)

Combined with the initial assumptions about the firm productivity process being lognormal, these results imply that the *conditional* distribution of the firm productivity process is $\ln(z) |\mu = \mu_b \sim \mathcal{N}(E(\ln(z) |\mu = \mu_b), Var(\ln(z) |\mu = \mu_b))$. And so I compute $\rho(\ln(z) |\mu = \mu_b)$ as:

$$\rho(z|\mu = \mu_b) = \rho(\ln(z)|\mu = \mu_b) = 2 \times \Phi\left(-\left|\frac{\ln(z) - E(\ln(z)|\mu = \mu_b)}{\sqrt{Var(\ln(z)|\mu = \mu_b)}}\right|\right)$$
(23)

where Φ is the standard normal cumulative distribution function. Expressions for $\rho(\mu = \mu_g)$, $\rho(z|\mu = \mu_g)$ and $\rho(\mu = \mu_g|z)$ are obtained similarly. In equilibrium, the probabilities $\rho(\mu = \mu_b|z)$ are consistent with the firm's decision rules, because they are based on the cross-sectional stationary distribution of firms $\Gamma(s)$, which is determined by the firm's decision rules. This means that the lender can price the firm-type risk such that the loan schedules are consistent with expected zero profits.

⁵In theory, $\ln(z)$ is a continuous random variable. Here I compute the empirical conditional mean and variance based on the discretized shock process, and so are using standard methods normally applied to compute the mean and variance of a discrete random variable.

3.6 The Cross-Sectional Distribution of Firms

The law of motion for the cross-sectional distribution of firms $\Gamma(s)$ is:

$$\Gamma(s') = \int \rho_d(s) Q_e(s') g(z') \Gamma(s) d(k \times l_R \times z \times \mu) + \int_{C(k', l'_R, \mu)} (1 - \rho_d(s)) Q_c(s, s') f(z', z) \Gamma(s) d(k \times l_R \times z \times \mu) + \int_{F(k', l'_R, \mu)} (1 - \rho_d(s)) Q_f(s, s') f(z', z) \Gamma(s) d(k \times l_R \times z \times \mu)$$
(24)

where $C(k', l'_R, \mu)$ is the continuing set, $F(k', l'_R, \mu)$ is the forbearance lending set, $\rho_d(k', l'_R, z, \mu)$ is the liquidation probability, and $Q_c(.)$ denotes a transition function for continuing firms, mapping the current states into future states, given by:

$$Q_{c}(s,s') = \begin{cases} 1 & \text{if } l'_{R}(s) = l'_{R}, k'(s) = k' \\ 0 & \text{otherwise} \end{cases}$$
(25)

and $l'_{R}(s)$ and k'(s) are the optimal policy rules for firms. Similarly, $Q_{f}(s')$ is 1 if the optimal choice for forbearance lending firms is (k', l'_{R}) and 0 otherwise. And $Q_{e}(s')$ is 1 if entrants optimal choice is (k', l'_{R}) and 0 otherwise.

If an existing firm liquidates, a new project z drawn from the stationary distribution g(z) becomes available to an entrant entrepreneur such that the mass of risky projects is equal to one.

3.7 Stationary Recursive Equilibrium

Definition 1. The stationary recursive equilibrium for this economy, where $s = \{k, l_R, z, \mu\}$ denotes the aggregate states and $V^*(s)$ is the firm's value function, is a set of policy rules for (i) capital holdings $k'^*(s)$, loan contracts $(l'^*(s), l'_R(s))$, and the default policy, where the default policy is comprised by the operating sets $C^*(k, l_R, \mu)$, liquidation sets $D^*(k, l_R, \mu)$, and forbearance lending sets $F^*(k, l_R, \mu)$; (ii) the recovery rate $\psi^*(s)$; (iii) the loan schedules $\Omega^*(k', z)$ for the loan contracts; and (iv) the cross-sectional distribution of firms $\Gamma^*(s)$ such that:

- 1. Given the loan schedule $\Omega^*(k', z)$ and the loan recovery rate $\psi^*(s)$: the value function $V^*(s)$, the policy rules $k'^*(s)$ and $(l'^*(s), l'_R(s))$, operating sets $C^*(k, l_R, \mu)$, liquidation sets $D^*(k, l_R, \mu)$, and forbearance lending sets $F^*(k, l_R, \mu)$ are consistent with the firm's optimization problem in Equation(3).
- 2. Given the loan schedule $\Omega^*(k', z)$ and value function $V^*(s)$: the loan recovery rate $\psi^*(s)$ solves the forbearance problem in Equation (13).

- 3. Given the loan recovery rate $\psi^*(s)$: the equilibrium loan schedules $\Omega^*(k', z)$ satisfy the lender's expected zero profit condition, and also the liquidation probability $\rho_d^*(k', l'_R, z, \mu)$, the forbearance probability $\rho_F^*(k', l'_R, z, \mu)$, the expected recovery rate $\gamma^*(k', l'_R, z, \mu)$, and the probability that a firm has permanent productivity μ_j given it has observed productivity $z \ \rho^*(\mu = \mu_j | z)$ are consistent with the firm's default policy and renegotiation agreement.
- 4. The cross-sectional distribution of firms $\Gamma^*(s)$ is a stationary measure of firms consistent with the firm decision rules and the law of motion for the stochastic variables.

4 Quantitative Analysis

4.1 Calibration

To consider the quantitative implications of the model, I set a model period to one year and calibrate the model to match euro-area data over the period 2011 to 2014. During this period, there was a high prevalence of forbearance lending (Acharya et al., 2019) and low output growth. Table 1 provides a summary of definitions used to compute key variables from the model.

I assume that firm productivity is a log-normal AR(1) process: $\ln (z_{jt}) = \mu_j (1 - \varphi) + \varphi \ln (z_{t-1}) + \epsilon$, with $|\varphi| < 1$ and $\epsilon \sim \mathcal{N}(0, \sigma_{\epsilon}^2)$. And that there are two firm types, low-quality firms and high-quality firms, that differ by the permanent productivity parameter μ_j . I use the procedure of Tauchen (1986) to discretize the stochastic shock into a 10-state Markov chain.

The model comprises 15 parameters to be calibrated, which are summarized in

Table 1. Model Delling	ions for Key variables
Variable	Model Definition
Assets	k
Loan repayment	l_R
Leverage	l_R/k
Loan interest rate	$l'_{R}/l' - 1$
Investment rate	$(k' - (1-\delta) k)/k^-$
Sales	zk^{lpha}
Sales growth	$\left(zk^{\alpha}-z^{-}\left(k^{-}\right)^{\alpha}\right)/z^{-}\left(k^{-}\right)^{\alpha}$
Dividend ratio	$I_{\{d>0\}}d/k^{-}$
Working capital	$\left(\hat{k} - \hat{l}_R ight) / k^-$

Table 1: Model Definitions for Key Variables

Table 2: Benchmark Parameters		
Calibrated Parameters		Value
Firm discount rate	β	0.96
Risk-free interest rate	r	0.01
Capital depreciation rate	δ	0.16
Returns to scale	α	0.65
Equity issuance cost	λ	0.30
Capital loss in liquidation	χ_d	0.40
Firm bargaining power	θ	0.40
Autocorrelation of stochastic shock		0.9
Standard deviation of stochastic shock	σ_{ϵ}	0.118
Low-quality firm permanent productivity		0
High-quality firm permanent productivity	μ_g	0.448
Parameters Estimated with SMM		Value
Capital adjustment cost	ϕ	0.15
Lender credit cost	ξ	7.00
Fixed operating cost	χ_c	10.75
Renegotiation cost	χ_f	0.20

Table 2. To start with, I select the 11 "Calibrated Parameters" in Table 2 independently of the model equilibrium. The firm discount rate β is set to 0.96, which is standard for an annual RBC model (Arellano, Bai and Zhang, 2012). The risk-free interest rate r is set to 0.01, which is equal to the average interest rate of one-year euro-area government bonds for the period 2011 to 2014. The returns to scale parameter α is set to 0.65, which implies decreasing returns to scale in production, similar as in the Arellano, Bai and Zhang (2012) study based on European firms. The equity issuance cost λ is set to 0.30, following Cooley and Quadrini (2001). The capital loss in liquidation χ_d and the firm bargaining power are both set to 0.40, which coincides with the estimates of Ramey and Shapiro (2001) for the capital loss in liquidation. The autocorrelation φ and standard deviation σ of the stochastic shock are set to 0.90 and 0.118, respectively, which follow ?.

Next, I use a sample of euro-area firms over the period 2011 to 2014 to estimate several other parameters. A full description of these data is contained in Appendix B. The capital depreciation rate δ is set to 16 per cent per year, which equals the average depreciation rate for the sample. To set the permanent productivity shocks, μ_b and μ_g , I divide the sample into low-quality firms and high-quality firms based on the definition of Acharya et al. (2019). This involves computing for each firm the average interest coverage ratio over the period 2009 to 2011, defined as the ratio of interest expenses to operating income (EBITDA). A firm is categorized as lowquality if its three-year average interest coverage ratio is below the country median and otherwise is considered high-quality. Using these categories, I find the average yearly return on assets for high-quality firms to be around 57 per cent higher than for low-quality firms. Thus, I set μ_b to zero and μ_g to 0.448, which ensures that the average productivity for high-quality firms is around 57 per cent higher than for low-quality firms for the specific productivity process.

Subsequently, I estimate the remaining four parameters in Table 2 by using the simulated method of moments (SMM) to target a set of five moment conditions from the data. More precisely:

$$\hat{\Theta} = \arg\min_{\Theta} \left[m^d - m^s(\Theta) \right]' W \left[m^d - m^s(\Theta) \right]$$
(26)

where W is a weighting matrix⁶, $\Theta = (\phi, \xi, \chi_c, \chi_r)$ are the four parameters to be estimated, m^d are the five targeted moments from the data summarized in the "Data" column of Table 3 and discussed below, and $m^s(\Theta)$ are the five simulated moments from the model at parameters Θ .

The first target moment is the firm liquidation rate. Standard & Poor's Rating Services (2014, 2016) provide an estimate for the yearly average of this moment for European corporates, which is 3.44 per cent.

The second target moment is the firm forbearance rate. The yearly average value of this moment over the period 2011 to 2014 is approximately 7.5 per cent, according to the estimates of Acharya et al. (2019). These estimates are based on the assetweighted proportion of European firms that are classified as "zombie firms". Their sample of firms comprises all European privately and publicly traded firms.

The remaining three target moments are: the average and standard deviation of sales growth, and the average of leverage. Appendix B describes how I compute these moments for the sample of euro-area firms.

All of the estimated parameters affect all of the target moments in the model, although some parameters affect some moments more directly. In particular, χ_c is useful to match the firm liquidation rate; χ_f is useful to match the firm forbearance lending rate; ϕ and χ_c is useful to match the average and standard deviation of sales growth; and ξ is useful to match average leverage.

I assess the quantitative performance of the model in Table 3 by comparing the moments from the euro-area data in the "Data" column with those produced by the model in the "Benchmark" column. Overall, the simulated method of moments

⁶I employ the identity matrix as the weighting matrix.

Targeted Moments	Data	Benchmark
Firm liquidation rate	3.44%	4.45%
Firm forbearance rate	7.5%	6.79%
Average of sales growth	5.86%	2.00%
Standard deviation of sales growth	16.62%	19.54%
Average of leverage	66.46%	68.69%
Non-targeted Moments	Data	Benchmark
Average of investment rate	30.47%	18.98%
Average of dividend ratio	3.48%	5.89%
Average of working capital	23.73%	29.99%

Table 3: Statistical Moments for Euro-Area Data and the Benchmark Model

Note: The column under Data reports the statistical moments for the euro area, from 2011 to 2014. The column under Benchmark reports the model statistics under the benchmark calibration. All statistical moments are calculated over a one-year period.

estimates a set of four parameter values that produce model moments that are close to the five targeted moments, although it under-predicts the average of sales growth. I also consider how the simulated model can match several non-targeted moments in Table 3. The model underestimates the investment rate and slightly overestimates working capital.

4.2 Loan Schedules and Firm Decisions

Prior to a discussion of the quantitative results, I first provide an overview of how liquidation risk and forbearance lending risk affect firms' loan schedules, and how these subsequently influence firms' choices about future loans and capital. I first consider how firm liquidation risk varies with firms' capital choice k' in Figure 1. Liquidation risk is higher for low-quality firms than for high-quality firms (Panel A). It is also higher for firms with less capital k, and for firms facing lower stochastic shocks (Panel B). In all three instances, the liquidation risk is higher because liquidation is more valuable.

I next consider how the firm forbearance lending risk varies as a function of the loan repayment choice l'_R in Figure 2. The forbearance lending risk is higher for firms with lower levels of physical capital, as well as for firms with larger loan repayment choices (Panel A). It is also higher for low-quality firms than for higher quality firms (Panel B) because high-quality firms are less likely to choose forbearance lending, even when the loan renegotiation generates some loan reduction. And although not depicted here, the forbearance lending risk is higher for firms facing lower stochastic shocks. As with liquidation, the forbearance lending option is more valuable for low-

quality firms, and for firms subject to lower stochastic shocks, as well as for firms with larger repayment choices.





Note: Both plots illustrate the liquidation probabilities as a function of the capital choice k' scaled by the average equilibrium capital \bar{k} . Panel A depicts two forms, one "high-quality" firm $(\mu = \mu_g)$ and one "low-quality" firm $(\mu = \mu_b)$, when there is a normal stochastic shock ε_m and the loan repayment choice l'_R equals the average equilibrium choice for all firms. Panel B depicts two firms, one with a low stochastic shock ε_l and one with a normal stochastic shock ε_m , when both firms are high-quality and the loan repayment choice l'_R equals the average equilibrium choice for all firms.





Note: Both plots illustrate the forbearance lending probabilities as a function of the loan repayment choice l'_R scaled by the average equilibrium capital \bar{k} . Panel A depicts two firms, one with low capital k_l and high capital k_h , where both firms are high-quality ($\mu = \mu_g$), face a normal stochastic shock ε_m and the capital choice k' is equal to the average equilibrium capital, \bar{k} . Panel B depicts two firms, one high-quality firm ($\mu = \mu_g$) and one low-quality firm ($\mu = \mu_b$), when there is a normal stochastic shock ε_m and the capital choice k' equals the median value.



Figure 3: Recovery Rate and Effective Interest Rate Schedule Note: Panel A plots the equilibrium recovery rates as a function of the loan repayment choice l'_R scaled by the average equilibrium capital \bar{k} . Panel B plots the effective interest rate $(l'_R/l'-1)$ for every contract (l', l'_R) as a function of the loan repayment choice l'_R scaled by the average equilibrium capital \bar{k} . Both plots depict two firms, one high-quality firm $(\mu = \mu_g)$ and one low-quality firm $(\mu = \mu_b)$, when there is a normal stochastic shock ε_m and the capital choice k' is equal to the average equilibrium capital \bar{k} .

Prior to a consideration of the loan schedules, I consider the loan recovery rates for forbearance lending. Panel A of Figure 3 demonstrates that a firm with a small loan repayment will receive no reduction to its loan. But the loan recovery rate decreases with the loan repayment size. The loan recovery rate is also higher for high-quality firms than for low-quality firms, all else equal. Similarly, and although not depicted here, the loan recovery rate is higher for firms with a more favorable stochastic shock. More generally, a larger loan reduction can increase a firm's *ex ante* forbearance incentives. But the lender anticipates this when pricing the loan schedule, which in turn offsets forbearance incentives.

I now consider the equilibrium loan schedules that arise due to liquidation risk, forbearance risk and firm-type risk. Panel B of Figure 3 illustrates that low-quality firms attract higher interest rates than high-quality firms. This is because they face both higher liquidation risk and forbearance lending risk than high-quality firms. Additionally, the effective interest rate increases with loan repayment size. But small loans also face higher effective interest rates due to the fixed cost of lending.

To consider the firm dynamics of the model, I turn to the policy rules. Panel A of Figure 4 shows that a firm with higher initial capital will choose a larger future capital stock and a smaller future loan. The smaller loan choice is due to a precautionary motive; the firm does not find it optimal to fully utilize its borrowing opportunities because large loans increase liquidation risk and forbearance risk, as well as the risk of





Note: Panel A plots the optimal capital choice k' as a function of initial capital k. Panel B plots the optimal loan choice relative to the capital choice l'/k' as a function of initial capital k. Both plots depict two firms, one high-quality firm ($\mu = \mu_g$) and one low-quality firm ($\mu = \mu_b$), when there is a high stochastic shock ε_h and the loan repayment l_R equals the average equilibrium choice for all firms. The horizontal axes and vertical axis of Panel A are scaled by the average equilibrium capital \bar{k} .

costly equity issuance (Arellano, Bai and Zhang, 2012). With lower levels of capital, the firm decreases its capital and increases its leverage (Panel B).

Loan contracts also influence the way in which firms respond to stochastic shocks. Arellano, Bai and Zhang (2012) note that when firms experience a sequence of low shocks, they will reduce their size in order to avoid equity issuance costs, and increase their loan financing, thus pushing up their leverage and their effective interest rate schedules. And visa versa for when firms experience a sequence of good shocks. Low-quality firms also tend to be smaller and more levered than high-quality firms because their liquidation risk is higher, and so they have more incentives to reduce scale in order to avoid equity issuance costs.

Panel B of Figure 4 shows that high-quality firms tend to be less levered than lowquality firms. In turn, lower leverage also reduces the liquidation risk and forbearance risk of high-quality firms.

4.3 Benchmark Model Results

I now examine the quantitative results for the benchmark model, organized by firm operating status, as reported in Table 4. All model moments are computed from a simulation of the model economy with 1,000 firms over 500 periods, having disregarded an initial 250 periods as a model burn-in.

I first consider the continuing firms in Column (2) of Table 4, which includes

	(1)	(2)	(3)	(4)	(5)
	All Firms	Continuing	For bearance	Liquidating	Entrants
Average of:					
Sales growth	2.00~%	1.07~%	-0.74~%	-10.75 %	15.18~%
Leverage	68.69~%	12.37~%	770.34%	250.86~%	149.26~%
Investment rate	18.98~%	17.65~%	14.17%	-38.82 %	38.40~%
Capital	126.52	139.32	64.42	43.26	27.62
Loan repayment	54.06	40.29	268.11	59.28	38.78
Firm age	46.72	52.89	12.76	22.47	2.37
TFP	1.74	1.79	1.41	0.98	1.63
Proportion of firms:					
In given category	100.00~%	88.77~%	6.79~%	4.45~%	$4.45 \ \%$
High-quality μ_a	90.28~%	99.65~%	0.13~%	50.02~%	54.69~%

Table 4: Quantitative Model Results by Operating Status

Note: Column (1) reports model statistics for all firms operating with age greater than one. Column (2) reports model statistics for continuing firms, defined as any firm that has existed for more than three periods and does not currently receive loan forbearance. Column (3) reports model statistics for forbearance firms, defined as any firm that has existed for more than three periods and currently receives loan forbearance. Column (4) reports model statistics for liquidating firms, based on the firm statistics in the period preceding firm liquidation. Column (5) reports model statistics for entrant firms, defined as any firm that has existed two periods or less. All model statistics are for the benchmark calibration, in which firms have access to forbearance lending.

all firms that have existed for more than three periods and currently do not receive loan forbearance. Almost all continuing firms are high-quality firms. One reason for this is because high-quality firms are more likely to have higher overall productivity, which makes these firms better able to make a successful entry. Continuing firms also tend to be larger with lower leverage due to precautionary motives, whereby these firms do not find it optimal to exhaust all borrowing opportunities due to the associated increase in liquidation risk and forbearance risk.

I next consider the firms in receipt of forbearance lending in Column (3) of Table 4, which includes all firms that have existed for more than three periods and currently receive loan forbearance. Only 0.13 per cent of firms that receive forbearance lending are high-quality firms. The reason that the majority of firms in receipt of forbearance lending are low-quality relates to the information asymmetry. Specifically, low-quality firms face a higher effective interest rate schedule than high-quality firms due to their higher liquidation and forbearance risk, and lower recovery rates (Figure 3). But the difference between the interest rate schedules of the high-quality firms and low-quality firms is not very large because the lender cannot perfectly predict whether a given firm is high-quality or low-quality, and must instead assign

a firm-type probability. The end result is that the interest rate schedule is more favorable for low-quality firms than it would be under perfect information, which makes forbearance lending *ex ante* a more valuable choice for low-quality firms.

The fact low-quality firms have lower overall productivity, on average, also increases their likelihood of receiving loan forbearance. This is because firms facing lower overall productivity, due to a bad stochastic shock or being low-quality or both, are more likely to opt for forbearance lending arrangements as the "continuing to operate" option is less valuable to them. The low overall productivity is consistent with the high level of leverage observed for the forbearance firms; when faced with a sequence of bad productivity shocks, firms will reduce their scale in order to avoid costly equity issuance, and increase their loan financing, thereby pushing up their leverage and effective interest rates. These model results are consistent with Banerjee and Hofmann (2020) who empirically show that in the years before a firm becomes a zombie, they increase leverage and equity issuance to stay afloat. Additionally, the model results are consistent with Goto and Wilbur (2019) and Bargagli-Dtoffi, Riccaboni and Rungi (2020) who empirically find that zombies are more prevalent among smaller firms.

Turning now to the result that firms in receipt of forbearance lending have lower sales growth, there are two factors that explain this. First, these firms are often experiencing a sequence of bad shocks, and this itself will lower sales growth. Second, these firms reduce their scale in response to the sequence of bad shocks, which further contributes to lower sales growth. Once more, these model results are consistent with the empirical findings in Banerjee and Hofmann (2020).

I now consider the liquidating firms in Column (4) of Table 4, which is based on statistics for the period preceding firm liquidation. These firms are equally likely to be high-quality or low-quality firms, although the path to liquidation differs by firm-type. High-quality firms tend to liquidate after being in the market for some time; their liquidation occurs if they experience a sequence of bad stochastic shocks. Low-quality firms, on the other hand, often liquidate after only a few periods only. When low-quality firms do survive for a longer period, this is due to their access to forbearance lending. For both firm-types, firms have high leverage and negative sales growth in the period before they liquidate, which is typical for any firm experiencing a sequence of bad stochastic shocks.

Finally, I consider the firm entrants in Column (5) of Table 4, which includes all firms that have existed for three periods or less. These firms are smaller and have higher sales growth rates, which is consistent with the findings of Arellano, Bai and Zhang (2012) for firm entrants.

In summary, the evidence in Table 4 suggests that nearly all firms in receipt of forbearance lending are low-quality, which is consistent with the empirical evidence of Acharya et al. (2019).

5 The Impact of Forbearance Lending

5.1 No Forbearance Lending Scenario

I use the calibrated model to examine the quantitative affects of forbearance lending via two counterfactual experiments. The first experiment aims to show how forbearance lending practices influence the financial decisions of firms, as well as firm dynamics and aggregate outcomes. To do this, I consider a counterfactual scenario in which the possibility of forbearance lending is shut down and so firms can only choose between continuing or liquidation. Here the value function of a firm that has the option to default and that starts the current period with capital k, a loan repayment l_R and productivity z is:

$$V(k, l_R, z, \mu) = \max_{\{c,d\}} \{ V_c(k, l_R, z, \mu), V_d(k, l_R) \},$$
(27)

where $V_c(k, l_R, z, \mu)$ is the value function if the entrepreneur does not default and continues to operate, as defined by Equation (4); and $V_d(k, l_R)$ is the value function if the entrepreneur defaults and liquidates the firm, as defined by Equation (5). The key departure from the benchmark model is that the value associated with forbearance, $V_f(k, l_R, z, \mu)$, is no longer an option in the value function.

In this counterfactual scenario, the equilibrium loan schedule $\Omega(k', z)$ that comprises all loan contracts (l', l'_R) allowing the lender to break even in expected values, and accounting for the liquidation risk that the lender faces, is described by:

$$l' = \sum_{j=b,g} \rho \left(\mu = \mu_j | z\right) \quad \left\{ \frac{1}{1+r} \left[1 - \rho_d \left(k', l'_R, z, \mu_j\right) \right] l'_R - \frac{1}{1+r} \rho_d \left(k', l'_R, z, \mu_j\right) \min\left\{ \chi_d k', l'_R \right\} \right\} - \xi$$
(28)

To generate the statistical moments for the counterfactual experiments, I employ the definitions outlined in Table 1. For the aggregate statistics, I compute aggregate investment, I, aggregate output, Y, aggregate capital, K, and aggregate total factor

	01100 101 0110	e o am corrace a da	Bulbourne
	(1)	(2)	(3)
Targeted Moments	Benchmark	No Forbearance	Perfect Information
Firm liquidation rate	4.45%	4.76%	5.14%
Firm forbearance rate	6.79%	0.00%	0.03%
Average of sales growth	2.00%	2.27%	3.92%
Standard deviation of sales growth	19.54%	19.40%	24.03%
Average of leverage	68.69%	18.55%	13.10%
Non-targeted Moments	Benchmark	No Forbearance	Perfect Information
Average of investment rate	18.98~%	19.69~%	25.41~%
Average of dividend ratio	5.89~%	5.57~%	6.44~%
Average of working capital	29.99%	78.21%	91.14%
Average of TFP	1.74	1.76	1.79
Aggregate investment	1.000	1.084	1.091
Aggregate TFP	1.000	1.008	1.016
Aggregate output	1.000	1.047	1.050

 Table 5: Statistical Moments for the Counterfactual Experiments

Note: The column under Data reports the statistical moments for the euro area, from 2011 to 2014. Column (1) reports model statistics under the benchmark calibration, in which firms have access to forbearance lending. Column (2) reports model statistics for the counterfactual experiment in which firms no longer have access to forbearance lending. Column (3) reports model statistics for the counterfactual experiment in which there is no information asymmetry and so lenders can observe firm-type. Aggregate statistics for output, investment and total factor productivity are scaled by the Benchmark Model statistics, and so equal to one for the Benchmark Model. All statistical moments and aggregates are calculated over a one-year period.

productivity, TFP, as follows:

$$I = \int i(s)\Gamma(s)d(k \times l_R \times z \times \mu)$$

$$Y = \int y(s)\Gamma(s)d(k \times l_R \times z \times \mu)$$

$$K = \int k(s)\Gamma(s)d(k \times l_R \times z \times \mu)$$

$$TFP = \frac{Y}{K^{\alpha}}$$
(29)

where $s = (k, l_R, z, \mu)$ are the states, $i = k' - (1 - \delta) k$ is firm investment, $y = zk^{\alpha}$ is firm output as measured by sales, and k is firm capital.

Table 5 presents statistical moments for the counterfactual experiments, in which Column (2) summarizes the results for the specific counterfactual experiment of no forbearance lending. Table 6 provides a more granular breakdown of the results by firm-type. On the basis of these results, I can conclude that with no forbearance lending: firm liquidation, average sales growth, aggregate total factor productivity (TFP), aggregate investment, and aggregate output are all higher, while average leverage is lower. I discuss these results in more detail below.

First, I focus on the liquidation risk and rate. Although not depicted here, I



Figure 5: Effective Interest Rate Schedules

Note: Panel A is for the benchmark model with forbearance lending. Panel B is for the counterfactual experiment with no forbearance lending. Panel C is for the counterfactual scenario with perfect information. All figures plot the effective interest rate $(l'_R/l'-1)$ as a function of the loan repayment choice l'_R (scaled by the average equilibrium capital \bar{k}). There are for two firms considered, one high-quality firm $(\mu = \mu_g)$ and one low-quality firm $(\mu = \mu_b)$, when there is a normal stochastic shock ε_m and the capital choice k' is equal to the average equilibrium capital \bar{k} .

find that the liquidation *risk* is significantly higher in the counterfactual experiment with no forbearance for otherwise identical firms. But the difference in the actual liquidation *rate* is not very significant (see Table 5). For example, the liquidation rate does not increase to the same extent that forbearance lending occurs - the other available default option - in the benchmark scenario. This is because in the absence of forbearance lending, firms make different equilibrium choices for their capital and loans, which offsets the observed increase in liquidation risk.

Next I consider the equilibrium loan schedules that arise in the counterfactual scenario of no forbearance lending. The equilibrium loan schedule is higher, on average, for firms in the counterfactual experiment than for the benchmark model (see Panel B in Figure 5). This is partly due to the higher liquidation risk of firms in the counterfactual scenario. But the more significant driver of the higher equilibrium loan schedule is the absence of forbearance lending. Forbearance lending risk is priced into the effective loan schedule. But unlike liquidation risk, forbearance lending risk can contribute to more favorable borrowing rates when the associated recovery rate is high because the lender is able to recuperate a much larger proportion of the loan repayment.

The loan schedule results also relate to the observations for overall average leverage. Table 5 shows that overall average leverage is around 73 per cent lower in the counterfactual experiment without forbearance lending. This is again due to the removal of the option for forbearance lending. The high leverage of low-quality firms in receipt of forbearance lending drives up the average leverage in the benchmark model. Additionally, the higher loan schedule in the counterfactual scenario of Panel B in Figure 5 disincentivizes firms to take on high levels of leverage.

Table 5 also shows that overall average sales growth is around 14 per cent higher in the counterfactual experiment without forbearance lending. The removal of the option of forbearance lending is a key driver of this result. Without access to forbearance lending, low-quality firms default soon after they enter. As such, there is a larger proportion of high-quality firms in the counterfactual scenario, which leads to higher average sales growth. Here low-quality firms in receipt of forbearance lending are no longer lowering the overall average sales growth, as in the benchmark model.

Similarly, Table 6 shows that aggregate investment is around 8 per cent higher in the counterfactual exercise as compared with the benchmark model, while aggregate output is around 5 per cent higher. Aggregate total factor productivity (TFP) is also a little higher. These differences in the aggregate statistics are, once more, due to the lower proportion of low-quality firms brought about by the removal of the option for forbearance lending.

These model results complement the empirical findings of Jorda et al. (2020) who show that for inefficient bankruptcy regimes, the pace of recovery following a corporate credit boom can be significantly dragged down.

These model results complement several empirical studies who show that zombie firm survival - sometimes connected to an inefficient bankruptcy regime - can impair aggregate productivity growth and the pace of recovery following recessions (see, e.g., Andrews and Petroulakis, 2019, Blattner, Farinha and Rebelo, 2018, Caballero, Hoshi and Kashyap, 2008, Jorda et al. (2020), Adalet McGowan, Andrews and Millot 2017, Adalet McGowan, Andrews and Millot, 2018).

It is worth noting that I find evidence of congestion effects in the model, consistent with the findings of Acharya et al. (2019) for European firms and Caballero, Hoshi and Kashyap (2008) for Japanese firms. This can be seen in Column (5) of Table 4 and Table 6, where the TFP for entrants is higher in the benchmark model with forbearance lending (Table 4) than in the counterfactual experiment with no forbearance lending (Panel A of Table 6). And in Column (4) of Table 4 and Table 6, where the TFP for liquidating firms is lower in the benchmark model with forbearance lending than in the counterfactual experiment with no forbearance lending. Together these results suggest that forbearance lending allows firms with lower productivity to continue operating and in so doing, restricts the prospects for new firms to enter. I do not, however, find evidence that the presence of "zombie firms" (firms in receipt of forbearance lending) harmed other "non-zombie firms", as documented by Acharya et al. (2019) and Caballero, Hoshi and Kashyap (2008). Instead, I find

	(1)	(2)	(3)	(4)	(5)
	All Firms	Continuing	Forbearance	Liquidating	Entrants
Panel A: Counterf	actual expe	riment with	no forbearan	nce	
Average of:					
Sales growth	2.27~%	1.03~%	n.a.	-3.34 %	18.36~%
Leverage	18.55~%	18.60~%	n.a.	17.59~%	17.95~%
Investment rate	19.69~%	17.59~%	n.a.	-17.77 %	46.84~%
Capital	134.54	142.68	n.a.	44.00	28.77
Loan repayment	54.28	57.76	n.a.	9.90	9.01
Firm age	50.00	53.68	n.a.	21.12	2.30
TFP	1.76	1.79	n.a.	1.05	1.47
Proportion of firms:					
In given category	100.00~%	95.24~%	n.a.	4.76~%	4.76~%
High-quality μ_g	96.94~%	99.79~%	n.a.	49.69~%	59.89~%
Panel B: Counterfa	actual expe	riment with	no informati	on asymmet	ry
Average of:					
Sales growth	3.92~%	1.87~%	-28.67~%	-5.56~%	109.44~%
Leverage	13.10~%	13.47~%	93.53~%	00.01~%	6.78~%
Investment rate	25.41~%	19.30~%	-38.65%	-32.74~%	352.33~%
Capital	133.96	138.74	262.93	27.05	24.34
Loan repayment	42.68	44.30	243.75	0.01	2.92
Firm age	51.98	54.00	41.45	19.47	2.50
TFP	1.79	1.79	1.46	0.99	1.82
Proportion of firms:					
In given category	100.00~%	94.83~%	0.03~%	5.14~%	5.14~%
High-quality μ_g	99.98~%	100.00~%	100.00~%	49.99~%	99.59~%

 Table 6: Quantitative Model Results for the Counterfactual Experiments

Note: Panel A reports model statistics for the counterfactual experiment in which firms no longer have access to forbearance lending. Panel B reports model statistics for the counterfactual experiment in which there is no information asymmetry and so lenders can observe firm-type. Column (1) reports model statistics for all firms operating with age greater than one. Column (2) reports model statistics for continuing firms, defined as any firm that has existed for more than three periods and does not currently receive loan forbearance. Column (3) reports model statistics for forbearance firms, defined as any firm that has existed for more than three periods and currently receives loan forbearance. Column (4) reports model statistics for liquidating firms, based on the firm statistics in the period preceding firm liquidation. Column (5) reports model statistics for entrant firms, defined as any firm that has existed two periods or less. All model statistics are for the benchmark calibration, in which firms have access to forbearance lending.

that the average sales growth, investment and productivity are not significantly different between the benchmark model and counterfactual scenario for continuing firms (Column (2) of Table 4 and Table 6), which is consistent with some of the findings of Schivardi, Sette and Tabellini (2017) and Schivardi, Sette and Tabellini (2020).

5.2 Perfect Information Scenario

I now examine the impact of information asymmetry on forbearance lending by conducting a counterfactual experiment with perfect information. With perfect information, the firm problem remains identical to the baseline scenario but the equilibrium loan schedule changes because the lender can perfectly identify firm type $j \in (b, g)$. The equilibrium loan schedule $\Omega(k', z)$ that comprises all loan contracts (l', l'_R) allowing the lender to break even in expected value is:

$$l' = \frac{1}{1+r} \left[1 - \rho_f \left(k', l'_R, z, \mu \right) - \rho_d \left(k', l'_R, z, \mu \right) \right] l'_R - \frac{1}{1+r} \left(\rho_f \left(k', l'_R, z, \mu \right) \cdot \gamma \left(k', l'_R, z, \mu \right) \right) l'_R - \frac{1}{1+r} \rho_d \left(k', l'_R, z, \mu \right) \min \left\{ \chi_d k', l'_R \right\} - \xi$$
(30)

Overall, Table 5 and Table 6 demonstrate that firm liquidation is higher, forbearance lending rates are lower, sales growth is higher, and leverage is lower under perfect information. I discuss these results in more detail below.

I first consider the impact of perfect information on the liquidation rate. Table 5 shows that the liquidation rate is about 16 per cent higher in the counterfactual experiment with perfect information than in the benchmark scenario with information asymmetry. The higher liquidation rate is mainly driven by higher defaults of entrants, as can be seen by the lower average age of liquidating firms in Table 4 and Table 6 (i.e. 22.47 versus 19.47). Although not depicted here, low-quality firms tend to have a higher liquidation risk in the counterfactual scenario with perfect information and so often liquidate after just one period. Under perfect information, lenders can properly discriminate between low-quality and high-quality firms when pricing loan schedules. The relatively higher loan schedules for low-quality firms reduces the relative value of continuing or forbearance lending, making liquidation typically the most valuable option for these firms.

Next I consider the equilibrium loan schedules that arise in the counterfactual scenario of perfect information. Compared with the benchmark scenario with information asymmetry, the loan schedule for high-quality firms is more favorable under perfect information, and vice versa for low-quality firms (see Panel C of Figure 5). This is because the lender can now perfectly identify firm-type and so more accurately price the loan schedule, and so the gap widens between the loan schedules of the high-quality and low-quality firms.

The loan schedules under perfect information lead to very different outcomes for the type of firms that obtain loan forbearance. With information asymmetry, almost exclusively low-quality firms obtain forbearance lending (see Table 4). With perfect information, only high-quality firms obtain forbearance lending because loan schedules under perfect information are priced such that low-quality firms do not find it optimal to choose forbearance lending (see Table 6). Yet, Table 6 also shows that the overall forbearance rate is much lower in the counterfactual experiment with perfect information. For the given model calibration, firms rarely find it optimal to choose forbearance lending under perfect information.

Average sales growth across all firms is around 96 per cent higher in the counterfactual experiment with perfect information than in the benchmark scenario, and also higher than in the counterfactual experiment with no forbearance lending (see Table 5 and Table 6). With perfect information, low-quality firms face high loan schedules because they have higher liquidation risk and forbearance risk. This contributes to these firms liquidating shortly after they enter. As such, the overall proportion of high-quality firms is 99.98 per cent under perfect information, which is higher than in both the benchmark and the counterfactual scenario of no forbearance lending. Once more, the consequence of a higher proportion of high-quality firms is that overall average sales growth is higher. For similar reasons, aggregate investment, aggregate output and aggregate TFP are respectively 9 per cent, 5 per cent and 2 per cent higher in the counterfactual scenario with perfect information as compared with the benchmark model.

6 Conclusions

In this study, I examine the relationship between forbearance lending, firms' financial decisions and firm dynamics. To do so, I develop a firm equilibrium model that features endogenous liquidations and endogenous forbearance lending. Lenders face information asymmetry because they do not know with certainty whether they are lending to a "low-quality" or "high-quality" firm. The model enables us to consider the net impact of forbearance lending on firm performance and aggregate outcomes by examining whether the costs of forbearance lending, due to credit misallocations from lenders incorrectly assessing a lender to be high-quality, outweigh its benefits, due to refinancing high-quality firms experiencing a negative sequence of shocks. I fit the model to the euro-area economy over the period 2011 to 2014, which represents a period of low output growth and higher levels of forbearance lending.

To examine the impact of forbearance lending, I conduct a counterfactual exercise in which firms still have the option to liquidate but no longer have access to any form of loan forbearance. In the absence of forbearance lending, I find that the average of firms' growth, investment rates and total factor productivity are higher in the counterfactual scenario with no forbearance lending. A key driver of this result is that there is a larger proportion of high-quality firms in the counterfactual scenario; low-quality firms are most likely to receive forbearance lending in the benchmark model, which weakens selection because it prevents these firms from liquidating more quickly. These low-quality firms, kept alive via forbearance lending, lower the aggregate output, investment and total factor productivity and they prevent new, more productive firms from entry.

The fact that the overwhelming majority of firms in receipt of forbearance lending in the model are of low-quality is consistent with the empirical evidence (see, e.g., Acharya et al., 2019). But while previous authors have emphasized the role of zombie lending incentives as a key driver of loan forbearance to low quality firms, my model highlights that information asymmetry faced by lenders is another factor. The results in this study also extend previous empirical findings by using a structural approach to quantifying the impact of forbearance lending on firm dynamics as well as various measures of firm performance and aggregate outcomes.

While the model provides new insights about the impact of forbearance lending on firms' financial decisions and their dynamics, some related questions remain unanswered. Specifically, the model does not include any role for the potential increased unemployment associated with higher firm liquidation rates that may occur in the absence of forbearance lending. A full examination of this potential benefit of forbearance lending presents an exciting avenue for future research. I also do not examine a role for forbearance lending incentives by weak banks, which the empirical literature finds to be important (see, e.g., Acharya et al., 2019, Andrews and Petroulakis, 2019, Blattner, Farinha and Rebelo, 2018, Giannetti and Simonov, 2013, Peek and Rosengren, 2005, Schivardi, Sette and Tabellini, 2017). Instead, I focus on the impact of these lending practices on the dynamics of firms, leaving related theoretical work on this topic open for future investigations.

In reality, I expect lenders to face information asymmetry. Therefore, my results suggest that the output-related costs of forbearance lending associated with misallocating credit to low-quality firms outweigh any output-related benefits associated with refinancing high-quality firms experiencing a sequence of negative shocks. From this perspective, the forbearance lending that has arisen since the onset of the sovereign debt crisis may have contributed the low output observed for the euro area over the same period.

References

- Acharya, Viral V., Lea Borchert, Maximilian Jager, and Sascha Steffen. 2020a. "Kicking the Can Down the Road: Government Interventions in the European Banking Sector." National Bureau of Economic Research NBER Working Paper 27537.
- Acharya, Viral V., Matteo Crosignani, Tim Eisert, and Christian Eufinger. 2020b. "Zombie Credit and (Dis-)Inflation: Evidence from Europe." National Bureau of Economic Research NBER Working Paper 27158.
- Acharya, Viral V, Tim Eisert, Christian Eufinger, and Christian Hirsch. 2019. "Whatever it takes: The real effects of unconventional monetary policy." The Review of Financial Studies, 32(9): 3366–3411.
- Adalet McGowan, Muge, Dan Andrews, and Valentine Millot. 2017. "Insolvency regimes, zombie firms and capital reallocation." OECD Working Paper 1399.
- Adalet McGowan, Muge, Dan Andrews, and Valentine Millot. 2018. "The walking dead? Zombie firms and productivity performance in OECD countries." *Economic Policy*, 33(96): 685–736.
- Andrews, Dan, and Filippos Petroulakis. 2019. "Breaking the shackles: Zombie firms, weak banks and depressed restructuring in Europe." European Central Bank Working Paper 2240.
- Arellano, Cristina, Yan Bai, and Jing Zhang. 2012. "Firm dynamics and financial development." Journal of Monetary Economics, 59(6): 533-549.
- Banerjee, Ryan Niladri, and Boris Hofmann. 2020. "Corporate zombies: Anatomy and life cycle." Bank for International Settlements Working Paper 882.
- Bargagli-Dtoffi, Falco J., Massimo Riccaboni, and Armando Rungi. 2020. "Machine Learning for Zombie Hunting. Firms Failures and Financial Constraints." IMT School for Advanced Studies Lucca Working Paper 01/2020.
- Bassetto, Marco, Marco Cagetti, and Mariacristina De Nardi. 2015. "Credit Crunches and Credit Allocation in a Model of Entrepreneurship." *Review of Economic Dynamics*, 18(1): 53–76.

- Blattner, Laura, Luisa Farinha, and Francisca Rebelo. 2018. "When losses turn into loans: the cost of undercapitalized banks." Banco de Portugal Working Paper w201816.
- Bonfim, Diana, Geraldo Cerqueiro, Hans Degryse, and Steven Ongena. 2020. "On-site inspecting zombie lending." CEPR Discussion Paper 14754.
- Caballero, Ricardo J., Takeo Hoshi, and Anil K. Kashyap. 2008. "Zombie Lending and Depressed Restructuring in Japan." American Economic Review, 98(5): 1943–1977.
- Cooley, Thomas F., and Vincenzo Quadrini. 2001. "Financial Markets and Firm Dynamics." *American Economic Review*, 91(5): 1286–1310.
- Corbae, Dean, and Erwan Quintin. 2015. "Leverage and the Foreclosure Crisis." Journal of Political Economy, 123(1): 1–65.
- Corbae, Dean, and Pablo D'Erasmo. 2021. "Reorganization Or Liquidation: Bankruptcy Choice And Firm Dynamics." *The Review of Economic Studies*.
- **D'Erasmo, Pablo.** 2011. "Government Reputation and Debt Repayment in Emerging Economies." Unpublished working paper.
- Giannetti, Mariassunta, and Andrei Simonov. 2013. "On the Real Effects of Bank Bailouts: Micro Evidence from Japan." American Economic Journal: Macroeconomics, 5(1): 135–167.
- Gopinath, Gita, Åebnem Kalemli-Ozcan, Loukas Karabarbounis, and Carolina Villegas-Sanchez. 2017. "Capital Allocation and Productivity in South Europe." The Quarterly Journal of Economics, 132(4): 1915–1967.
- Goto, Yasuo, and Scott Wilbur. 2019. "Unfinished business: Zombie firms among SME in Japan's lost decades." Japan and the World Economy, 49(C): 105–112.
- Hennessy, Christopher A., and Toni Whited. 2007. "How Costly Is External Financing? Evidence from a Structural Estimation." *Journal of Finance*, 62(4): 1705–1745.
- Hopenhayn, Hugo, and Richard Rogerson. 1993. "Job Turnover and Policy Evaluation: A General Equilibrium Analysis." Journal of Political Economy, 101(5): 915–938.

- Hsieh, Chang-Tai, and Pete Klenow. 2009. "Misallocation and Manufacturing TFP in China and India." The Quarterly Journal of Economics, 124(4): 1403–1448.
- Jorda, Oscar, Martin Kornejew, Moritz Schularick, and Alan M. Taylor. 2020. "Zombies at Large? Corporate Debt Overhang and the Macroeconomy." National Bureau of Economic Research NBER Working Paper 28197.
- Khan, Aubhik, and Julia K. Thomas. 2013. "Credit Shocks and Aggregate Fluctuations in an Economy with Production Heterogeneity." Journal of Political Economy, 121(6): 1055–1107.
- Kwon, Hyeog Ug, Futoshi Narita, and Machiko Narita. 2015. "Resource Reallocation and Zombie Lending in Japan in the 1990s." *Review of Economic Dynamics*, 18(4): 709–732.
- Midrigan, Virgiliu, and Daniel Yi Xu. 2014. "Finance and Misallocation: Evidence from Plant-Level Data." *American Economic Review*, 104(2): 422–458.
- Moll, Benjamin. 2014. "Productivity Losses from Financial Frictions: Can Self-Financing Undo Capital Misallocation?" American Economic Review, 104(10): 3186-3221.
- Peek, Joe, and Eric S. Rosengren. 2005. "Unnatural Selection: Perverse Incentives and the Misallocation of Credit in Japan." American Economic Review, 95(4): 1144–1166.
- Ramey, Valerie A., and Matthew D. Shapiro. 2001. "Displaced Capital: A Study of Aerospace Plant Closings." *Journal of Political Economy*, 109(5): 958– 992.
- Restuccia, Diego, and Richard Rogerson. 2008. "Policy Distortions and Aggregate Productivity with Heterogeneous Plants." *Review of Economic Dynamics*, 11(4): 707–720.
- Schivardi, Fabiano, Enrico Sette, and Guido Tabellini. 2017. "Credit misallocation during the European financial crisis." Bank of Italy Working Paper 1139.
- Schivardi, Fabiano, Enrico Sette, and Guido Tabellini. 2020. "Identifying the Real Effects of Zombie Lending." The Review of Corporate Finance Studies, 9(3): 569-592.

- Senkal, Asil. 2014. "Misallocation and Recovery Rates." Unpublished working paper.
- Standard & Poor's Rating Services. 2014. "Oil on Troubled Waters: European Corporate Credit Outlook Q4 2014."
- Standard & Poor's Rating Services. 2016. "Recovery, Divergence and Disruption: European Corporate Credit Outlook 2016."
- Storz, Manuela, Michael Koetter, Ralph Setzer, and Andreas Westphal. 2017. "Do we want these two to tango? On zombie firms and stressed banks in Europe." European Central Bank Working Paper 2104.
- **Tamayo, Cesar.** 2017. "Bankruptcy Choice with Endogenous Financial Constraints." *Review of Economic Dynamics*, 26: 225–242.
- Tauchen, George. 1986. "Finite state markov-chain approximations to univariate and vector autoregressions." *Economics Letters*, 20(2): 177–181.
- Yue, Vivian Z. 2010. "Sovereign default and debt renegotiation." Journal of International Economics, 80(2): 176–187.

A Computational Algorithm

I first set grids on the state space of the aggregate states of the economy, $s = \{k, l_R, z, \mu\}$. The productivity shock z is discretized into 10-state Markov chain using the method of Tauchen (1986). The state space of the loan repayment amount l_R and capital k are each discretized into a grid of 15 and 25 points, respectively, between 0 and 400. The loan repayment amount grid and the capital grid are unevenly spaced, using a similar method to Corbae and Quintin (2015). That is, every point of an equally spaced grid between 0 and $400^{2/3}$ is raised to the 3/2 power. As in Corbae and Quintin (2015), this grid contains more points closer to zero and so provides improved numerical performance for our problem.

I solve the model to find the optimal loan schedules, policy functions and default policies using the following algorithm:

- 1. For all aggregate states of the economy, $s = \{k, l_R, z, \mu\}$, start with an initial guess for the bank loan recovery schedule ψ^0 . For all k' and z, start with an initial guess for the loan schedule Ω^{00} , in which all loan contracts (l', l'_R) have the risk-free interest rate. For the probability of firm-type given observed productivity $q^0 = \Pr(\mu = \mu_j | z)$, start with an initial guess by using $\Pr(\mu = \mu_b) = \Pr(\mu = \mu_g) = 0.5$ and $\Pr(z | \mu = \mu_b) = 2 \times \Phi\left(-\left|\frac{\ln(z) \mu_b}{\sigma_{\varphi}}\right|\right)$.
- 2. Given the bank loan recovery schedule ψ^0 and the equilibrium loan schedule Ω^0 , use value function iterations to solve for the optimal policy functions for future capital stock k'(s), the new bank loan contract $(l'(s), l'_R(s))$, operating sets $C(k, l_R, \mu)$, liquidation sets $D(k, l_R, \mu)$, and forbearance sets $F(k, l_R, \mu)$. I iterate on the value function until I reach convergence for a given ψ^0 and Ω^0 .
- Given the operating sets, liquidation sets, forbearance sets and q⁰: compute the new debt schedule Ω¹ such that lenders break even in expectation, and compare it to loan debt schedule of the previous iteration: Ω⁰. If a convergence criterion is met, max {Ω⁰ Ω¹} < ε, then assign Ω¹ to Ω⁰ and move on to the step 4. Otherwise, update using a Gauss-Seidel algorithm and go back to step 2.
- 4. Solve the bargaining problem given the converged loan schedule Ω^0 and compute the new bank loan recovery schedule ψ^1 for all aggregate states of the economy, $s = \{k, l_R, z, \mu\}$. If the new bank loan recovery schedule ψ^1 is sufficiently close to ψ^0 , stop iterating on ψ . Otherwise, go back to step 2.

- 5. To compute the stationary cross-sectional distribution, start with a uniform distribution as an initial guess. Then simulate the stationary cross-sectional distribution associated with the set of policy functions obtained above, and such that the mass of all projects always equals one. This implies that the mass of entrants equals the the mass of liquidations. Update q^1 based on $\Pr(\mu = \mu_b)$ and $\Pr(z|\mu = \mu_b)$ from the stationary cross-sectional distribution. If the firm-type probability given productivity q^1 is sufficiently close to q^0 , stop iterating on q. Otherwise, go back to step 2.
- 6. Simulate the model to compute statistics from a model economy of 1,000 firms over 750 periods, using the first 250 periods as a burn-in.

B Data and Parameter Estimation

I use firm-level data to estimate for three parameters in Table 2 independently of the model equilibrium, as well as to estimate most of the model statistics listed in Table 3. I obtain our firm-level data from the Amadeus database, which comprises financial data for both public and private companies in Europe. Our sample comprises all public and private euro-area firms, excluding the financial and government sectors. I collect annual financial data for the period 2009 to 2014.

I clean the dataset in several ways. Following Arellano, Bai and Zhang (2012), I exclude firms in the financial and government sectors. Financial sector firms correspond to NACE Divisions 64 to 67 and government sector firms correspond to NACE Division 84. I furthermore collect data for euro-area firms only. The sample is also restricted to those firms with a reported value for EBIT (that is, earnings before interest and taxes), sales, and fixed assets in at least one of the years from 2012 to 2014. These criteria leave us with around 150,000 firms in 17 countries: all euro-area countries except for Ireland. Table 7 describes how I compute the parameters and model statistics.

Calibrated Parameters	Definition
Depreciation	$\frac{\text{Depreciation}_t}{\text{Fixed Assets}_{t-1}}$
Interest Coverage Ratio	$\frac{\text{EBIT}_t}{\text{EBIT}_t - \text{EBT}_t}$
Model Statistics	Definition
Sales Growth	$\frac{\text{Sales}_{t} - \text{Sales}_{t-1}}{\text{Sales}_{t-1}}$
Leverage	$\frac{\text{Total Assets}_t - \text{Total Equity}_t}{\text{Total Assets}_t}$
Working Capital	$\frac{\text{Working Capital}_{t}}{\text{Total Assets}_{t-1}}$
Investment	$\frac{\text{Fixed Assets}_{t} - \bar{\text{Fixed Assets}}_{t-1} + \text{Depreciation}_{t-1}}{\text{Fixed Assets}_{t-1}}$

 Table 7: Definitions of Parameters and Model Statistics from Firm-level Data

 Calibrated Parameters
 Definition

Note: Each variable is windsorized at the 5 per cent level. The model statistics are computed for each firm in each year from 2011 to 2014. I first take the mean and standard deviation of each statistic in each year. The model statistics are then calculated as the mean of the yearly mean (for the mean of the model statistic) or the mean of the yearly standard deviation (for the standard deviation of the model statistic) over the period 2011 to 2014.